Author's response to reviews

Title: Effect of vaginal self-sampling on cervical cancer screening rates: a community-based study in Newfoundland.

Authors:

Pauline Duke (pduke@mun.ca)
Marshall Godwin (godwinm@mun.ca)
Samuel Ratnam (sam.ratnam@easternhealth.ca)
Lesa Dawson (lmdawson@mun.ca)
Dan Fontaine (hickeyfontaine@hotmail.com)
Adrian Lear (adrian.lear@easternhealth.ca)
Martha Traverso-Yepez (mtraverso@mun.ca)
Wendy Graham (wendy.graham@med.mun.ca)
Mohamad Ravalia (ravalia@mun.ca)
Gerry Mugford (gmugford@mun.ca)
Andrea Pike (andrea.pike@med.mun.ca)
Jacqueline H Fortier (jacqueline.fortier@med.mun.ca)
Mandy Peach (c64mep@mun.ca)

Version: 4
Date: 18 January 2015

Author's response to reviews:

The size of the cohorts has been included in the abstract as requested.
A footnote has been added to Table 1 describing the communities in more detail.